Reviewer # 1
Specific comments to authors
Very nicely done trial and so well written too. please check table 2 for completeness of AGI grading in heading.

ANSWER: Table 2 has been perfected, which has been marked in yellow in the page 19.

Caption for fig is inserted below the fig and that for table is inserted above it.

ANSWER: Have been perfected, which has been marked in yellow in the page 17-21.

Please elaborate ROC curve findings and provide interpretation for ROC AUC.

ANSWER: The article has been explained, which has been marked in yellow in the page 8.

Reviewer # 2
Specific comments to authors
The title should be changed as "Heparanase inhibition improves patients with acute gastrointestinal injuries induced by sepsis".

ANSWER: The title has been revised as required, but the retouching expert suggested that it be changed to "Heparanase inhibition leads to improvement in patients with acute gastrointestinal injuries induced by sepsis", which has been marked in yellow in the article.

Please clarify the figure 1 regarding the patients who loss to follow. Because these sepsis patients were admitted in ICU during the study.

ANSWER: The article has been explained, which has been marked in yellow in the page 7.

For Table 1, please recheck for the last column (P-value) and the caption again. For Figure 3, please delete "as well as". The reference format is not correct.
ANSWER: Have been modified, which has been marked in yellow in the page 17-19. The reference format is not correct.

The English grammar needs to be polished.

The reference format have been modified, and the article has been retouched. The new language qualification certificate has been uploaded.